Title: Induction of Vitiligo after Imiquimod Treatment of condyloma acuminatum

Authors:

hongyan xin (13789823558@163.com)
lingshi ge (463496488@qq.com)
hanyan song (sh2130@163.com)
wang cao (279871020@qq.com)
wenfei li (lw8888026@163.com)

Version: 3 Date: 31 May 2014

Author's response to reviews:

Cover Letter
Dear editor:
According to your request, I have made some changes in my manuscript.
1. Version: 2 Date: 27 March 2014 Reviewer: Ian Denham Reviewer's report:
Minor essential revisions:
The abstract is too long and needs to be condensed. The plural form ie condylomata acuminata (genital warts) should be used throughout
Yes, I have condensed the abstract, and the number of words is changed from 324 to 168.
All of Condyloma acuminatum (venereal wart) in my manuscript have changed into condylomata acuminata (genital warts).
2. Quality of written English: Needs some language corrections before being published.
Yes, I asked International Science Editing Ltd to help me in English.
3. Version: 2 Date: 8 May 2014 Reviewer: Marcus Chen Reviewer's report: This is a lengthy report for a case report. I would suggest that it is shortened.
Some of the details can be left out such as the values for the laboratory investigations.
The word limit should conform to the word limit for a case report for the journal.
Yes, the abstract has been shorted. And the values for the laboratory investigations are left out.
4. I am not sure that I agree with the conclusion that: "when prescribing this drug, dermatologists must take measures to prevent this possible side effect.
Given it is a very rare side effect I am not sure discussion of this is warranted in all cases. Also how can it be prevented?
Yes, “This adverse effect should be considered during treatment. When
prescribing this medication, the dermatologist should take measures to prevent this side effect.” in abstract has been changed into “This adverse effect should be considered when dermatologists prescribe this medication.”

“Among its adverse effects, imiquimod-induced vitiligo should be anticipated. When prescribing this drug, dermatologists must take measures to prevent this possible side effect.” in conclusion in main body of my manuscript has been changed into “Among its adverse effects, imiquimod-induced vitiligo should be anticipated when dermatologists prescribe this drug.”

5. In order to make my revised manuscript conforms to the journal style, I have added a title to each figure.

Thank you and best regards.

Yours sincerely,
Wenfei Li, MD PHD
Corresponding author:
Wenfei Li
Department of Dermatology, Qianfoshan Hospital, Shandong University, Jinan, Shandong, 250014, China.
lwf888026@163.com